Government 320: Public Opinion and Public Choice

Spring 2006 Tuesday and Thursday 2:55–4:10 (MG 165) Professor: Walter R. Mebane, Jr.

Office: 217 White Hall (255-3868); email wrm1@cornell.edu Office hours: T 3:00–5:00, W 3–4 or other times by appointment. Course web page:

http://macht.arts.cornell.edu/wrm1/gov320.html

• is election fraud a kind of heresthetic?

– dimension manipulation (maybe) – strategic voting (maybe) – agenda control (yes) • does Arrow’s Theorem legitimate election fraud?